Author's response to reviews

Title: Randomised trials relevant to mental health conducted in low and middle-income countries: survey of studies published in 1991, 1995 and 2000 and assessment of their relevance

Authors:

Rebecca J Syed Sheriff (rebecca_syled@hotmail.com)
Mahesh Jayaram (maheshbj@gmail.com)
Prathap Tharyan (prathap@cmcvellore.ac.in)
Lelia Duley (lelia.duley@ndm.ox.ac.uk)
Clive E Adams (ceadams@cochrane-sz.org)

Version: 3 Date: 21 September 2006

Author's response to reviews: see over
To Whom It May Concern,

The following minor changes have been made to this revision of the document

-I have changed the formatting as requested

-my name, I recently got married

-the methods for the identification of false negatives has been improved by the decision to manually search all reports in English rather than only a 10% simple.

-there have been minor changes to the proposed analysis due to advice from Colin Mathers at the WHO that the burden of disease statistics for the sample years were not suitable for conducting a time series.